Author's response to reviews

Title: Single blind randomized Phase III trial to investigate the benefit of a focal lesion ablative microboost in prostate cancer (FLAME-trial): study protocol for a randomized controlled trial

Authors:

Irene M. Lips (i.m.lips@umcutrecht.nl)
Uulke A. van der Heide (u.vd.heide@nki.nl)
Karin Haustermans (karin.haustermans@uzleuven.be)
Emile van Lin (e.vanLin@rther.umcn.nl)
Floris Pos (f.pos@nki.nl)
Stefan P.G. Franken (s.fraken-3@umcutrecht.nl)
Alexis N.T.J. Kotte (akotte@umcutrecht.nl)
Carla H. van Gils (c.vangils@umcutrecht.nl)
Marco van Vulpen (m.vanvulpen@umcutrecht.nl)

Version: 5 Date: 25 November 2011

Author's response to reviews:

I hereby send you the revised manuscript. the following changes were made:

* title is identical in the manuscript and the submission system
* the author list is identical in the manuscript and the submission system
* the abstract is identical in the manuscript and the submission system
* the conclusion section has been amalgamate into the discussion section
* the references are cited in the text using square brackets

Please contact me if you have any questions or other changes,

Best wishes,

Irene lips